Author’s response to reviews

Title: The Application of Virtual Reality and Augmented Reality in Oral & Maxillofacial Surgery

Authors:

Ashraf Ayoub (Ashraf.Ayoub@glasgow.ac.uk)

Yeshwanth Pulijala (yesh@myscalpel.com)

Version: 2 Date: 27 Aug 2019

Author’s response to reviews:

Based on the reviewers’ comments, we have carried out the following corrections:
1. Edited the title as suggested
2. Edited the abstract as recommended
3. Edited the introduction as suggested
4. Added the two extra manuscripts as required
5. Expanded the literature search and added more articles (41-46) to the manuscript
6. Focused and edited the conclusion section
7. Reformatted the citation of the references
8. Reformatted the references.